Author's response to reviews

Title: Randomised social-skills training and parental training plus standard treatment versus standard treatment of children with attention deficit hyperactivity disorder - The SOSTRA trial protocol.

Authors:

Ole Jakob Storebø OJS (ojst@regionsjaelland.dk)
Jesper Pedersen JP (jpee@regionsjaelland.dk)
Maria Skoog MS (maria.skoog@ctu.rh.dk)
Per Hove Thomsen PHT (per.hove.thomsen@ps.rm.dk)
Christian Gluud CG (cgluud@ctu.rh.dk)
Erik Simonsen ES (es@regionsjaelland.dk)

Version: 2 Date: 12 November 2010

Author's response to reviews:

We have included a short discussion section in the abstract, included figure titles after the reference list in the manuscript, and finally we have added copies of the original funding and ethics documentation.